Author's response to reviews

Title: Association of Khat and alcohol use with HIV infection and age at first sexual initiation among youths in Gamo-Gofa Zone, South west Ethiopia

Authors:

Marelign Malaju Tilahun Mr. (marikum74@Gmail.com)
Gistane Ayele Asale Miss. (gistanea@yahoo.com)

Version: 2 Date: 13 September 2012

Author's response to reviews:

This research article is useful for developing countries like Ethiopia in the reduction of HIV prevalence where the prevalence is still high because of early sexual initiation and substance abuse and still the determinant factors of early sexual initiation and HIV infection in Ethiopia is not well studied especially in the rural setting.

Therefore, this article will be helpful for countries like Ethiopia and other researchers to give attention for reduction of HIV prevalence among youths.